Author's response to reviews

**Title:** A Cross-Curricular Physical Activity Intervention to Combat Cardiovascular Disease Risk Factors in 11-14 Year Olds: 'Activity Knowledge Circuit'

**Authors:**

Gareth Knox (gaknox@uwic.ac.uk)
Julien S Baker (Julien.Baker@uws.ac.uk)
Bruce Davies (bdavies1@glam.ac.uk)
Susan Faulkner (sfaulkne@glam.ac.uk)
Jaynie Rance (J.Y.Rance@swansea.ac.uk)
Anwen Rees (anrees@uwic.ac.uk)
Kelly Morgan (kelly_morgan1987@hotmail.co.uk)
Non Thomas (N.E.Thomas@swansea.ac.uk)

**Version:** 2 **Date:** 25 November 2009

**Author's response to reviews:** see over
The Editor,
BMC Public Health,

Dear Editor,

Please find below point-by-point description of changes made to the manuscript entitled ‘A Cross-Curricular Physical Activity Intervention to Combat Cardiovascular Disease Risk Factors in 11-14 Year Olds: ‘Active Learning Pathway’.

- Within the title ‘Active Learning Pathway’ has been changed to ‘Activity Knowledge Circuit’.

- In the Method/Design section under Intervention ‘Active Learning Pathway’ has been changed to ‘Activity Knowledge Circuit’.

- The trail registration number has been added at the end of the abstract.

- Within the Method/Design section under Data Analysis ‘A Pre- Post-intervention design’ has been changed to ‘A quasi-experimental design’. A power calculation has been included along with a description of dependent and independent T-tests.

Yours Faithfully,

Gareth Knox